Author's response to reviews

Title: A Distal Region of the Human TGM1 Promoter Is Required for Expression in Transgenic Mice and Cultured Keratinocytes

Authors:

Marjorie A Phillips (maphillips@ucdavis.edu)
Bart A Jessen (bart.jessen@pfizer.com)
Ying Lu (yilu@ucdavis.edu)
Qin Qin (qaqin@ucdavis.edu)
Mary E Stevens (mary.stevens.b@bayer.com)
Dr Robert H Rice (rhrice@ucdavis.edu)

Version: 3 Date: 14 Mar 2004

March 13, 2004

Dear Editor,

Thanks for your email of March 12 informing us that our manuscript 1321115670229900 "A Distal Region of the Human TGM1 Promoter Is Required for Expression in Transgenic Mice and Cultured Keratinocytes" is provisionally accepted for publication. We have read the reviewers reports and note that they did not recommend any further changes. We have made the requested formatting changes, corrected several typographical errors and added one further person to the acknowledgments.

Sincerely,

Robert H. Rice
Professor